RESEARCH

Open Access



Psychometric properties of the nine-item avoidant/restrictive food intake disorder screen (NIAS) in Turkish children

Hakan Öğütlü^{1*}¹, Meryem Kaşak², Uğur Doğan³, Hana F. Zickgraf^{4,5,6} and Mehmet Hakan Türkçapar⁷

Abstract

Background The nine item avoidant/restrictive food intake disorder screen (NIAS) is a short and practical assessment tool specific to ARFID with three ARFID phenotypes such as "Picky eating," "Fear," and "Appetite". This study aimed to evaluate the psychometric properties of the Turkish translation of the NIAS parent form and to investigate the relationship between ARFID symptoms and anxiety, depression symptoms, and eating behaviors in a sample of Turkish children.

Method Parents were asked to provide their children's sociodemographic data and to complete the NIAS, Eating Disorder Examination Questionnaire-Short (EDE-QS), Children's Eating Behavior Questionnaire (CEBQ), and Revised Child Anxiety and Depression Scale (RCADS) scales.

Results The sample included 440 participants between 6 and 12 ages. Turkish NIAS demonstrated good internal consistency. The three-factor model of the Turkish NIAS was in an acceptable structure. The Turkish NIAS scale was shown to be valid and reliable. NIAS scores were shown to be higher in underweight participants. The NIAS-parent version subscales showed expected convergent and divergent validity with the CEBQ, EDEQ-S, and RCADS scales in children, except CEBQ emotional overeating and desire to drink subscales were correlated with NIAS.

Conclusion The Turkish version of the NIAS is valid and reliable in evaluating ARFID symptoms in children.

Keywords NIAS, ARFID, Avoidant/restrictive food intake disorder, Validation, Children

Plain English summary

Assessment tools for avoidant/restrictive food intake disorder (ARFID), which is quite common in children, are quite limited. NIAS-parent version is a practical and valuable scale that can be used in the clinic. This study found that the Turkish version of the NIAS is valid and reliable in evaluating ARFID symptoms in children. In the psychometric properties of the Turkish NIAS, ARFID symptoms were associated with anxiety and depression symptoms and food-avoidant eating behavior. It was also found that ARFID was not associated with eating disorder symptoms. In addition, the paper shows initial data concerning the psychometric properties related to the Turkish NIAS-parent version. It is the first study to evaluate the relationship of ARFID subtypes with anxiety, depression symptoms, appetite characteristics, and BMI percentages in children.

*Correspondence: Hakan Öğütlü hogutlu@gmail.com Full list of author information is available at the end of the article



© The Author(s) 2024. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.gr/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.gr/licenses/by/4.0/. The Creative Commons Public Domain and the credit line to the data.

Background

Avoidant/restrictive food intake disorder (ARFID) is characterized by food avoidance or dietary restriction that causes significant weight loss, growth retardation in childhood, dependence on nutritional supplements (i.e., oral or enteral formulas), or impairment in psychosocial functioning [1]. Although ARFID was not a new condition and has been observed by clinicians for a long time, it was first introduced to psychiatric nosology with the publication of DSM-5 in 2013 [1]. It is worth noting that prior to being labeled as ARFID, clinicians referred to similar conditions using different terms, including but not limited to infantile anorexia and feeding disorder of infancy and early childhood [2].

Thomas et al. [3] proposed a three-dimensional model of neurobiological abnormalities in sensory perception, homeostatic appetite regulation, and negative valence systems that underlie the three essential aspects of ARFID: selective/neophobic eating, lack of interest in foods and eating, and fear of aversive consequences, respectively. Parental inability to model appropriately (for example, no family meals, skipping meals, watching TV during meals, parents' disordered eating symptoms), parents' authoritarian or permissive feeding practices, and limited exposure to food may play a role as environmental risk factors [4, 5]. However, parental behaviors have not been extensively studied in ARFID, and limited longitudinal data are available to assess the direction of the relationship between familial factors and child ARFID symptoms.

It is essential to detect ARFID, which causes significant psychosocial and medical problems, in the clinic, but screening tools are minimal and not specific to diagnosing ARFID. Cooney et al. [6] stated that psychometric measurement tools widely used to evaluate eating disorders are not sensitive and specific for diagnosing ARFID in the pediatric group.

The NIAS is a short and practical assessment tool measuring the degree to which each ARFID phenotype, such as "Picky eating," "Fear," and "Appetite" is experienced [7]. A parent-report version of the NIAS for children ages five and up has been developed and is used clinically. Still, to our knowledge, only the selective eating subscale has been used in published empirical research [8, 9]. The NIAS scale has been translated into multiple languages [10] after being validated in a large population sample in the United States [7], but no Turkish validity study was performed. In addition, as far as is known, no study comprehensively evaluated the relationship of all three ARFID phenotypes with anxiety, depression symptoms, and eating behaviors in children using the NIAS.

Therefore, this study aimed to evaluate the psychometric properties (factor structure, reliability, and convergent and divergent validity) of the Turkish version of the NIAS parent form, to determine the validity of the scale in Turkish children, and to investigate the relationship between ARFID symptoms and body weight (body mass index percentile), weight/shape-related disordered eating, anxiety, and depression symptoms and appetitive traits linked to body weight in childhood. Based on patterns observed using the adult self-report NIAS (e.g., [7]) we predicted that the picky eating and appetite subscales would be negatively related to appetitive traits associated with food approach, including food responsiveness and enjoyment of eating. Based on findings that ARFID is more commonly comorbid with anxiety than depression, we predicted that all subscales would be related to overall anxiety. Finally, we predicted that the NIAS subscales would be uncorrelated with a measure of weight/ shape-related disordered eating symptoms and that the appetite subscale would be negatively correlated with the BMI percentile [7].

Methodology

Participants

A population sample of children aged 6–12 was used in the study. There were no exclusion criteria in the study. The study sample was taken from primary and secondary schools (i.e., 1st to 7th grades) in Muğla, Turkey. In order to prevent selection bias, the school and class selections of the participants in the study were provided by the Muğla Guidance and Research Center using the blind method in selecting the study group; the non-random, convenient sampling method was adopted. Of the 734 participants in the selected classes, 266 (36.2%) parents did not consent to the study, and 28 (3.8%) participated but did not complete all of the scale items. The final sample included 440 participants, 217 (49.31%) girls and 223 boys (50.68%), with a mean age of 9.08 ± 1.9 years. Mothers (n = 382, 86.8%), fathers (n = 44, 10%), and other relatives (n = 14, 3.18%) completed the ratings. The mean weight of the children was 32.74 kg (±10.55), height was 134.45 cm (±14.04), Body Mass Index (BMI) was 17.73 (±3.35), and percentiles were 58.1 (± 33.22).

Upon evaluating the cut-off scores of children using the Revised Child Anxiety and Depression Scale (RCADS), it was found that 92.0% of the sample (405 children) were clinically normal for depression, while 8% (35 children) were at the clinical level. In terms of anxiety, 80.9% (356 children) were classified as normal, and 19.1% (84 children) were at the clinical level. The breakdown of clinical levels for different anxiety subtypes is as follows: Generalized Anxiety—16.4% (72 children) clinical, Panic—18.9% (83 children) clinical, Separation Anxiety—23.9% (105 children) clinical, Social Phobia—14.5% (64 children) clinical, and Obsessions/Compulsions—18.9% (83 children) clinical, with 76.1% (335 children) being normal.

For test–retest reliability, 70 randomly selected participants were asked to re-answer the NIAS one month after the initial administration. Fifty-two participants (74.2%) answered the items again.

Procedure

Permission was obtained from the developer of NIAS to adapt the scale to Turkish children via e-mail. Ethics committee approval was obtained for the study from the Social Sciences University of Ankara Institutional Ethics Committee of Social Sciences and Humanities Research and Publication (No:2022/44352, Date: 08/08/2022). Then, in the first stage, the English version of the NIAS was translated into Turkish by two Turkish clinical psychologists with a good command of English. These two psychologists translated the scales independently, followed by consensus after a discussion between the two translators and the corresponding authors of this article. Next, a professional bilingual translator who had no previous knowledge of the scale translated the Turkish form of the scale back into English. The NIAS developer later reviewed the translated version, and some minor wording changes were made. Afterward, the final version of the scale was applied to the parents of 10 children, the researchers corrected misunderstandings, and the Turkish version was given its final form.

Next, an online survey was created using the Survey Monkey website (www.surveymonkey.com) to be administered to the participants. On the first page of the questionnaire, the content and objectives of the study were detailed, and all the necessary information for informed consent was given to the parents. Informed consent was obtained from the participants. In the questionnaire, parents were asked to provide their children's socio-demographic (i.e., age, gender), weight, and height data and to complete the NIAS, Eating Disorder Examination- Questionnaire-parent (EDE-QS-parent), Children's Eating Behavior Questionnaire (CEBQ), and RCADS scales.

Measurements

Nine item avoidant/restrictive food intake disorder screen (NIAS)

Developed by Zickgraf & Ellis (2018) [7], the NIAS is a 9-item self-report scale that evaluates avoidant/restrictive eating disorder symptoms. The NIAS has three subscales: Picky eating (items 1–3), Appetite (items 4–6), and Fear (items 7–9). Each item of the NIAS is scored between 0 ('strongly disagree') and 5 ('strongly agree'). Each subscale is scored on a scale of 0–15, with higher scores indicating higher levels of each subscale. All items can also be added together to calculate an overall score ranging from 0 to 45, with higher scores indicating higher levels of avoidant/restrictive eating overall. In the original study, Cronbach's α value for the NIAS total score to assess internal reliability was 0.90 [7]. In this study, Cronbach's α was 0.81 for the NIAS total score.

Eating disorder examination-questionnaire short (EDE-QS)

The EDE-QS is a 12-item single-factor scale developed by Gideon et al. [11] that evaluates the core symptoms of AN, BN, and eating disorder not otherwise specified (EDNOS). It is the abbreviated form of EDE-Q [12]. Each item of the EDE-QS is scored between 0 (0 days/ Not at all) to 3 (6–7 days/Markedly), and a total score is obtained by summing and averaging the items; higher scores indicate more severe levels of eating disorders. In the original study, Cronbach's α value for the EDE-QS total score was 0.91. The validity and reliability study of the Turkish version of this scale in an adult sample was conducted by Esin et al. [13]. In this study, Cronbach's α value for the Turkish version of the parent version of EDE-QS total score was 0.88.

Children's eating behavior questionnaire (CEBQ)

This scale was developed by Wardle et al. [14] to determine children's eating behaviors. The scale, answered by the parents, consists of 35 items. Each item of the CEBQ is scored between 1 ('never') and 5 ('always'). The CEBQ has eight subscales: food responsiveness (FR), emotional overeating (EOE), enjoyment of food (EF), desire for drinks (DD), satiety responsiveness (SR), slowness in eating (SE), emotional undereating (EUE), and food fussiness (FF). Yılmaz et al. [15] carried out the Turkish validity of the scale. The Cronbach's α values for the CEBQ ranged from 0.70 to 0.89 in the original study [14].

Revised child anxiety and depression scale (RCADS)

RCADS was developed by Chorpita et al. [16] to screen for anxiety disorders and depression in children and adolescents. The scale, which has two versions as child and parent forms, consists of 47 items. Each item of the RCADS is scored between 0 ('never') and 3 ('always'). The RCADS has six subscales: generalized anxiety disorder (GAD) (6 items), separation anxiety disorder (SPAD) (9 items), social anxiety disorder) (SAD) (7 items), panic disorder (PD) (9 items), obsessive–compulsive disorder (OCD) (6 items), and major depressive disorder (MDD) (10 items). Görmez et al. [17] carried out the Turkish validity of the scale. The Cronbach's α values for the RCADS ranged from 0.78 for SAD to 0.88 for GAD in the original study [16].

Body mass index (BMI)

BMI was calculated (kg/m2), and the BMI percentiles for age were determined using World Health Organisation (WHO) growth charts. Based on the recommended cutoff points of BMI percentiles for Turkish children and adolescents [18] (i.e., <5 percentiles = underweight, 5–85 percentiles = normal weight, >85 percentiles = overweight). Parents of the participants reported their children's weight and height, which were used to compute BMI. The BMI values of the participants were used in the study because they were related to the A.1 criterion (weight loss due to restrictive eating) of ARFID diagnosis according to DSM-5.

Data analysis

Statistical analyses of scale adaptation were made with the JASP (2020) [19] program. JASP is software built on the R (R Core Team, 2019) [20] program, using R packages [21]. The student's t-test and ANOVA were performed for the genders and BMI percentile categories of the subscale scores and total scores of the NIAS. In cases where significant differences were detected, post hoc analysis (Bonferroni) was performed to determine the source of the difference between the groups.

Confirmatory factor analysis (CFA) was used in the data analysis to test the construct validity. For the CFA analysis at scale, the JASP program uses the "lavaan" [22], "semPlot" [23], and "psych" [21] packages based on the CFA analysis assumptions in Brown (2014) [24] and Kline's (2015) (35) books. For CFA, one of the estimation methods, "Maximum likelihood" was used. The studied data meet the three basic assumptions of Maximum likelihood: (1) the sample is large, (2) the data is continuous, and (3) it requires multivariate normally distributed indicators. Fit indices were evaluated

as a result of CFA, according to recommended values for an adequate model fit based on the literature: Chisquared statistic/degrees of freedom (χ 2/df) < 5, rootmean-square error of approximation (RMSEA) < 0.08; Goodness-of-Fit Index (GFI) > 0.90 [25], Comparative Fit Index (CFI) \geq 0.95, Standardized Root Mean Square Residual (SRMR) \leq 0.08 [26, 27], Tucker–Lewis index (TLI) \geq 0.90 [26].

Pearson correlation test was used to examine the relationship between "Picky eating" (NIAS-picky eating), "Fear" (NIAS-fear), and "Appetite" (NIAS-appetite) subscale scores of NIAS, EDE-QS, CEBQ, and RCADS scores, and children's BMIs. In addition, partial correlation analysis was conducted to control variables.

Item-total correlation, test–retest, and Cronbach's α internal consistency coefficient were used for the reliability analysis. Item-total correlation and test–retest (initial and follow-up scores of NIAS) were performed with Pearson's correlation test. The mean values of items were expressed with standard deviation, and results with p < 0.05 are considered statistically significant.

Results

Factor analysis of the NIAS

The averages of the items of the NIAS ranged from 0.61 to 2.89. The means and standard deviations of the items on the scale are shown in Table 1.

Construct validity

Confirmatory factor analysis (CFA)

First-level CFA was performed to determine the construct validity to adapt the NIAS scale to the Turkish language. It was seen that the three-dimensional

Table 1 Descriptive statistics, adjusted item-total correlation and Cronbach's α value of NIAS

Dimension	Items	Mean	SD	Item Total Correlation	lIDCα	t-test	Cα
Picky eating	NIAS 1	2.89	1.54	0.55	0.79	- 17.49*	0.84
	NIAS 2	2.00	1.53	0.59	0.78	-19.42*	
	NIAS 3	1.85	1.62	0.60	0.78	- 20.35*	
Appetite	NIAS 4	1.63	1.53	0.63	0.78	- 18.73*	0.81
	NIAS 5	2.11	1.60	0.57	0.79	-20.01*	
	NIAS 6	1.28	1.37	0.45	0.80	-13.26*	
Fear	NIAS 7	0.61	1.02	0.40	0.81	-8.22*	0.87
	NIAS 8	0.70	1.07	0.41	0.81	-8.53*	
	NIAS 9	0.63	0.99	0.37	0.81	-8.40*	
Full scale							0.81

NIAS Nine Item Avoidant/Restrictive Food Intake Disorder Screen, SD Standard Deviation, IIDCα If Item Dropped Cronbach's α; t-test for Distinguishing Features of Items, Cα Cronbach's α

t Test**p* < 0.05

model of the measurement tool was confirmed in the CFA result of the NIAS scale. When the fit indices related to the analysis were examined, it was seen that it gave excellent results ($\chi 2 = 59.3$, df = 24, $\chi 2/df = 2.47$; RMSEA = 0.058 [Cl lower = 0.039, Cl upper = 0.076]; CFI = 0,981; TLI = 0,971; SRMR = 0.037). The standardized factor loading values were 0.74 and 0.91 on the whole scale. It was seen that all factor loads are significant, and the residual covariances of the items are not high. Factor loadings and the CFA results of the scale are given in Table 2.

For the CFA, the covariances between the factors were calculated, and the results are given in Table 3. It was seen that the covariance between NIAS-picky eating and NIAS-appetite is 0.55, the covariance between NIAS-fear and NIAS-picky eating is 0.20, and the covariance between NIAS-appetite and NIAS-fear is 0.24.

Descriptive analysis of the NIAS

First, the dimensions and total scores of NIAS were defined and compared according to the BMI percentile category and gender. Specifically, for BMI, underweight participants showed significantly higher scores on the NIAS full-scale score and NIAS-appetite than those in the healthy weight and overweight/obese range (all p < 0.05). There were no significant gender differences in any subscale or the total, score. The results are shown in Table 4.

Criterion validity

The criterion validity of the NIAS was assessed according to the correlation between subscale scores and BMI percentile values, CEBQ, EDE-QS, and RCADS scores.

	Table 2	Factor	loadings	and CFA	results	of NIAS
--	---------	--------	----------	---------	---------	---------

Table 3 Factor covariance of NIAS

	Est	SE	95% Cl	95% CI	
			Lower	Upper	
Picky eating					
Picky eating	1				
Appetite	0.55***	0.04	0.46	0.63	
Fear	0.20***	0.05	0.10	0.31	
Appetite					
Appetite	1				
Fear	0.24***	0.05	0.14	0.34	
Fear					
Fear	1				

NIAS Nine Item Avoidant/Restrictive Food Intake Disorder Screen, SE Standard Eror, CI Confidence Interval

^{***} p < 0.001

There was no significant relationship between NIAS total score and EDE-QS (r=0.06, p=0.22). There was a negative and significant relationship between the NIAS total score and BMI (r=-0.17, p<0.001) of the children. Finally, there were positive and significant relationships between the NIAS total score and the RCADS all anxiety sub-dimensions scores, the total anxiety score (r=0.23, p<0.001), and the depression score (r=0.34, p<0.001).

Table 5 shows the correlation results of the three NIAS subscales with each variable. While a small-moderate and negative relationship existed between BMI percentile and NIAS-appetite, there was no significant relationship between other subscales and BMI. NIAS-appetite and NIAS-fear showed a small and significant correlation with parent-reported weight/shape eating disorder symptoms on the EDEQ-S. In contrast, NIAS-picky eating had no relationship with non-ARFID eating disorder symptoms. Overall, NIAS-appetite showed the expected

	ltems	ltems Estimate	SE	SE %95 CI		St. Est	RC
				Lower	Upper		
Picky eating	NIAS 1	1.14*	0.06	1.00	1.27	0.74	0.45
	NIAS 2	1.28*	0.06	1.15	1.40	0.84	0.30
	NIAS 3	1.35*	0.06	1.22	1.49	0.84	0.30
Appetite	NIAS 4	1.24*	0.06	1.11	1.37	0.81	0.34
	NIAS 5	1.36*	0.07	1.22	1.49	0.85	0.28
	NIAS 6	0.87*	0.06	0.75	0.10	0.64	0.59
Fear	NIAS 7	0.81*	0.04	0.73	0.89	0.80	0.36
	NIAS 8	0.85*	0.04	0.76	0.94	0.79	0.37
	NIAS 9	0.90*	0.04	0.82	0.98	0.91	0.17

NIAS Nine Item Avoidant/Restrictive Food Intake Disorder Screen, SE Standard Eror, CI Confidence Interval; RC Residual Covariances, St.Est: Standart Estimate *p<0.05

	Underweight (n = 97) M (SD)	Normal (n = 193) M (SD)	Overweight (n = 150) M (SD)	F	Post-Hoc
Picky eating	7.10 (±4.18)	6.77 (± 3.67)	6.53 (±4.36)	0.63	_
Appetite	6.22 (±4.04)	5.42 (±4.18)	4.10 (±3.53)	11.58*	UW > N > OW
Fear	2.03 (±3.06)	1.96 (±2.65)	1.92 (±2.74)	0.05	-
NIAS total	15.35 (±8.80)	14.15 (±7.69)	12.55 (±7.45)	4.45*	UW > N > OW
	(5irls n = 217) Λ (SD)	Boys (n = 223) M (SD)		t
Picky eating	6	5.76 (±4.09)	6.72 (±4.11)		0.10
Appetite	5	5.38 (± 3.93)	4.67 (± 3.71)		1.95
Fear	1	.94 (±2.79)	1.97 (±2.78)		-0.12
NIAS total	1	4.08 (±7.91)	13.36 (±7.90)		0.95

Table 4	Descriptive A	Analysis and	Group Com	parisons of NIAS I	by BMI and Gender
---------	---------------	--------------	-----------	--------------------	-------------------

NIAS Nine Item Avoidant/Restrictive Food Intake Disorder Screen, BMI Body Mass Index, SD Standard Deviation, UW Underweight, N Normal, OW Overweight * p < 0.05

pattern of relationships with the CEBQ subscales, with positive and strong relationships with SR and SE and moderate-large negative relationships with EF and FR. The relationships between NIAS-Appetite and EUE/EOE were small-moderate but significant and in the expected direction, and there was a small but significant positive relationship with FF. Also, as predicted, NIAS-picky eating was most strongly correlated with the CEBQ food fussiness scale. The relationships between NIAS-picky and the other CEBQ subscales mirrored those of NIASappeite, with generally smaller effect sizes. The exceptions were a null relationship with food responsiveness (FR) and a strong relationship with satiety responsiveness (SR). NIAS-fear had small but significant positive associations with EOE, DD, SR, SE, and EUE and a null relationship with FR and FF. All three subscales were positively associated with overall Depression and Anxiety symptoms on the RCADS; effect sizes for NIAS-picky eating and NIAS-appetite were small, whereas the relationships with NIAS-fear were moderate. Of the RCADS symptom subscales, NIAS-picky eating had the strongest relationship (r=0.31) with OCD symptoms. In contrast, NIAS-fear had the strongest relationships with panic disorder (r=0.33) and MDD (r=0.33). All three subscales had small, positive relationships with SAD symptoms. Still, only Fear was significantly associated with GAD, with a small effect size.

Partial correlation analysis was performed to control the EDEQ-S score for the relationship between NIAS scores and BMI percentile, CEBQ subscales, and RCADS scores. Again, while a negative and significant relationship was found between BMI percent and NIAS-appetite (r = -0.27, p < 0.001), no significant relationship was found between other subscales and BMI (p > 0.05). When symptoms of eating disorders other than ARFID were controlled, the negative and significant relationship between NIAS-appetite and CEBQ's food approach subscales (FR, EOE, and EF) and the positive and significant relationship with food avoidance subscales (SR, SE, and EUE) continued. NIAS-picky eating was also positively and strongly associated with FF (r=0.49, p < 0.001). Again, all three subscales remained positively associated with overall Depression and Anxiety symptoms on the RCADS (p < 0.001) (Table 5).

Reliability analysis of the NIAS *Internal consistency*

The internal consistency of the scale was tested by using Cronbach's α value. Cronbach's α was 0.81 for the NIAS total score, 0.84 for Picky Eating, 0.81 for Appetite, and 0.87 for Fear. It can be said that the values between 0.80 and 0.90 that emerged as a result of the analyses have "good reliability" [28].

Distinguishing features of items

Another way to ensure reliability is to compare the lower 27% and upper 27% groups. Since the lower 27% and upper 27% groups formed according to the total scores obtained from the measurement tool are expected to be different in terms of the measured feature, there is expected to be a significant difference between the item average scores of the groups. The t-test was conducted to determine the significance of the differences between the item average scores of the upper 27% (N:118) and lower 27% (N:118) groups from the study population. According to the t-test results of the groups, it was found that there was a significant (p < 0.001) difference between the lower and upper groups. Regarding this result, the items

Table 5 Convergent and divergent validity of the NIAS

Scales	NIAS	Pearson's r	p	Pearson's r(Controlling for EDE-QS)	p
EDE-QS	Picky_Eating	0.04	0.389	-	_
	Appetite	-0.11*	0.028	_	-
	Fear	0.25***	< 0.001	_	_
BMI Percantil	Picky Eating	-0.07	0.160	-0.08	0.105
	Appetite	-0.26***	< 0.001	-0.24***	< 0.001
	Fear	-0.03	0.477	-0.09	0.062
CEBO					
Food responsiveness	Picky Fating	-0.05	0.316	-0.07	0.169
Food responsiveness	Appetite	-0.38***	< 0.001	-0.37***	< 0.001
	Fear	0.05	0 303	-0.02	0.731
Emotional overeating	Picky Fating	-0.02	0.747	-0.04	0.444
Emotional overeating	Annetite	-0.29***	< 0.001	-0.27***	< 0.001
	Fear	0.12**	0.003	0.09	0.064
Enjoyment of food	Picky Eating	_037***	< 0.001	-0.38***	< 0.001
Enjoyment or lood		-065***	< 0.001	-065***	< 0.001
	Foar	-0.11*	0.032	- 0.14**	0.004
Dosiro to drink	Picky Esting	0.11*	0.032	0.11*	0.004
Desire to utilik	Appetito	-0.01	0.825	0.01	0.031
	Appente	- 0.01	0.825	0.09	0.945
Satisty responsiveness	Dicky Esting	0.11	< 0.023	0.00	< 0.092
satiety responsiveness	PICKy_Eduling	0.55	< 0.001	0.55	< 0.001
	Appente	0.55	< 0.001	0.55	< 0.001
Classical static statics of	Fear Dislos Estima	0.10**	0.004	0.1/^^^	< 0.001
Slowness in eating	PICKy_Eating	0.29***	< 0.001	0.29***	< 0.001
	Appetite	0.4/***	< 0.001	0.47***	< 0.001
	Fear	0.12**	0.004	0.18***	< 0.001
Emotional undereating	Picky_Eating	0.19***	< 0.001	0.18***	< 0.001
	Appetite	0.1/***	< 0.001	0.19***	< 0.001
	Fear	0.13**	0.007	0.10*	0.036
Food fussiness	Picky_Eating	0.48***	< 0.001	0.49***	< 0.001
	Appetite	0.19*	0.028	0.29***	< 0.001
	Fear	0.01	0.821	0.14**	0.004
RCADS					
SAD	Picky_Eating	0.17***	< 0.001	0.17**	0.001
	Appetite	0.18***	< 0.001	0.19***	< 0.001
	Fear	0.16**	0.003	0.13*	0.010
GAD	Picky_Eating	0.08	0.109	0.06	0.219
	Appetite	0.10	0.060	0.11	0.055
	Fear	0.22***	< 0.001	0.14**	0.009
MDD	Picky_Eating	0.22***	< 0.001	0.22***	< 0.001
	Appetite	0.23***	< 0.001	0.30***	< 0.001
	Fear	0.33***	< 0.001	0.25***	< 0.001
PD	Picky_Eating	0.04	0.489	0.01	0.851
	Appetite	0.14**	0.009	0.19***	< 0.001
	Fear	0.31***	< 0.001	0.23***	< 0.001
SPAD	Picky_Eating	0.11*	0.036	0.09	0.079
	Appetite	0.09	0.085	0.13*	0.010
	Fear	0.23***	< 0.001	0.15**	0.005

Scales	NIAS	Pearson's r	p	Pearson's r(Controlling for EDE-QS)	р
OCD	Picky_Eating	0.31***	< 0.001	0.25***	< 0.001
	Appetite	0.10	0.066	0.14*	0.055
	Fear	0.07	0.178	0.05	0.338
Depression	Picky_Eating	0.22***	< 0.001	0.22***	< 0.001
	Appetite	0.23***	< 0.001	0.30***	< 0.001
	Fear	0.33***	< 0.001	0.25***	< 0.001
Anxiety Total	Picky_Eating	0.12*	0.021	0.10*	0.048
	Appetite	0.15**	0.005	0.20***	< 0.001
	Fear	0.29***	< 0.001	0.21***	< 0.001
Total	Picky_Eating	0.15**	0.004	0.14**	0.009
	Appetite	0.17***	< 0.001	0.23***	< 0.001
	Fear	0.31***	< 0.001	0.23***	< 0.001

Table 5 (continued)

NIAS Nine Item Avoidant/Restrictive Food Intake Disorder Screen, EDE-QS Eating Disorder Examination- Questionnaire Short, BMI Body Mass Index, CEBQ Children's Eating Behaviour Questionnaire, RCADS Revised Child Anxiety and Depression Scale, SAD Social Anxiety Disorder, GAD Generalized Anxiety Disorder, MDD Major Depressive Disorder, PD Panic Disorder, SPAD Seperation Anxiety Disorder, OCD Obsessive Compulsive Disorder, Total Total Anxiety&Depression

 $p^* < 0.05, p^{**} < 0.01, p^{***} < 0.001$



Fig. 1 Correlation heatmap of items of the NIAS scale

have good discrimination. The t values for the analysis are given in Table 1.

Item analysis

The item-total correlation results of the scale ranged from 0.55 to 0.60 for Picky eating, 0.45 to 0.63 for Appetite, and 0.37 to 0.41 for Fear. The fact that the items had high correlations with the scale scores and each other indicates a potential hierarchical latent structure. Values related to the analysis are given in Table 1.

In Fig. 1, the heat map turns blue if there is a positive relationship between the variables, and the heat map turns red if there is a negative relationship. If there is no significant relationship between the variables, it does not take on any color, and there is a cross (X) over the cell (r=0.07). Except for the insignificant relationship between the first and ninth items of the NIAS scale, the other items were significantly related to each other. The heat map for the correlation analysis is given in Fig. 1.

Test-retest

Correlation analysis was performed to determine the relationship between the two administrations. Statistically significant and strong associations were found for Picky eating (r=0.94, p < 0.001), Appetite (r=0.94, p < 0.001), Fear (r=0.82, p < 0.001), and the total score (r=0.95, p < 0.001). A very high similarity was found between the two applications in terms of the total score and sub-dimensions of the scale.

Discussion

In this study, the validity of the parent form of the Turkish NIAS scale in children was investigated, and its psychometric properties were examined. The Turkish NIAS scale was shown to be valid and reliable. Through the reliability analysis, the NIAS demonstrated good internal consistency. The three-factor model of the Turkish NIAS was found to be in an acceptable structure by CFA analysis. NIAS scores were shown to be higher in underweight participants. The NIAS-parent version subscales showed expected convergent and divergent validity with the CEBQ, EDEQ-S, and RCADS scales in children.

In the study that the NIAS was developed and first validated, Cronbach's α for the Total score was 0.90 [7], 0.86 in the Chinese NIAS validity study [10], 0.88 in the Polish study [29], 0.84 in the Mexican study [30], and alpha 0.81 in the Turkish NIAS. In addition, the three-factor structure in our study also exhibited internal consistency similar to other studies.

In addition, the three-factor NIAS model was acceptable in the CFA analysis. NIAS-picky eating, NIASappetite, and NIAS-fear subscales of the goodness of fit indices were compatible. This model supports the separation of ARFID in the DSM-5 as in the original study [7]. In the scale, NIAS-picky eating indicated the presentation of selective/neophobic ARFID, NIAS-appetite the presentation of lack of interest, and NIAS-fear the presentation of fear of negative consequences [31].

In factor covariances and correlations between NIAS subscales, Picky eating, Appetite, and Fear subscale scores were found to be correlated with each other. These relationships show that the scale harmoniously evaluates the three presentations to be measured. Good inter-item correlation coefficients show that the three subscales measure the ARFID presentation in a standard way [7]. In addition, as in the studies conducted in the USA (adults), China (college undergraduates), and Mexico (adolescents), no gender differences were found in the NIAS total score and subscale scores [10, 30].

In our study, only NIAS-appetite was associated with a low BMI percentile, and underweight children had higher scores on this subscale. Still, no such relationship could be shown between NIAS-picky eating and NIAS-fear. The results for Turkey, located at the intersection of Europe and Asia, were in line with the literature on the relationship between selective eating and body weight in Western countries, as expected [USA (e.g., [7, 32], United Kingdom (e.g., [33]), and Australia (e.g., [34]]. In China, NIAS-appetite and NIAS-picky eating were associated with lower BMI in young adults, which the authors speculated was due to a less obesogenic food environment in China compared with the US, Australia, and Europe [10]. The literature for the USA/Europe/Australia is somewhat mixed, but systematic reviews suggest no relationship between children's picky eating and BMI in these countries [35]. Understandably, being less motivated to eat and restricting volume (i.e., Appetite ARFID symptoms) is protective against obesity in most countries, but restricting variety may only be protective against obesity in certain food environments. In highly obesogenic food environments such as the USA and Mexico, there is sometimes a slight positive correlation between picky eating and BMI [36]. Picky eaters, particularly young children, may indeed prefer certain processed foods with high calories. These foods are often highly palatable due to their high fat, sugar, and salt content, making them more appealing to picky eaters [37].

As hypothesized, there was evidence of divergent validity with the EDEQ-S, a measure of eating disorder symptoms maintained by cognitive restraint, weight and shape concerns, and fear of weight gain [11]. The presence of a small positive correlation between NIAS-picky eating and the EDEQ-S is consistent with findings from the adult NIAS. NIAS validation studies from treatmentseeking samples with eating disorders suggest that while the NIAS is valid and reliable in this population, it has a relatively poor ability to discriminate between ARIFD and other EDs due to the tendency of patients with non-ARFID EDs to endorse ARFID symptoms on the NIAS [31]. As a result, it is recommended to use NIAS and EDE-Q together to evaluate both ARFID and other eating disorders for a more complete picture of the eating behaviors and motivations behind them than is offered by either measure alone [31].

On the other hand, the CEBQ is a scale with different subscales that measures appetitive traits, early-emerging physiological, emotional, and behavioral responses to food and eating that are linked to weight gain (food approach traits) or protection against obesity (food avoidance traits) [15]. The CEBQ measures traits that may represent risk factors for ARFID. As hypothesized, NIAS-appetite showed positive correlations with food avoidance subscales and negative correlations with food approach subscales of the CEBQ. This relationship of NIAS-appetite with Appetite's physiological, motivational, emotional, and behavioral dimensions shows that it is consistent with the ARFID symptoms defined in DSM-5 [1]. NIAS-picky eating was negatively associated with enjoyment of food, while it was positively associated with satiety responsiveness, slowness in eating, emotional undereating, and especially food fussiness [38]. When EAT-26 scores were controlled, the direction and significance of the relationship between NIAS-appetite and CEBQ food approach and food avoidance did not change. Similarly, NIAS-picky eating and FF maintained a positive and strongly associated relationship. As in the adult validation study, the NIAS-fear subscale had weak or null relationships with appetitive traits. Whereas picky eating and appetite disturbances are early-emerging traits often first identified before age 5 in patients who develop selective and Appetite ARFID symptoms, fear ARFID has an acute onset often associated with a conditioning event like choking or vomiting [39, 40]. Temperamental risk factors for this ARFID presentation are likely to overlap with risk for anxiety and affective disorders [3].

The rates of psychiatric disorders accompanying ARFID are high, ranging between 57 and 95% [6, 41]. Anxiety disorders are most common in 36–72% [6, 42], and generalized anxiety disorder is the most common comorbid anxiety disorder in youth with ARFID (although this may reflect the base rate of GAD being higher than that of other anxiety disorders) [41, 43]. Mood disorders accompany ARFID with the second frequency between 17 and 33% [42, 44]. In the relationship with RCADS scores, there was a positive relationship between NIAS-appetite, NIAS-picky eating, NIAS-fear, total anxiety, and depression scores. When non-ARFID eating disorder symptoms were controlled, the relationship between Appetite, Picky eating and Fear subscales, and anxiety and depression measures did not change. Picky eating frequently contributes to the symptoms in clinical samples of those diagnosed with ARFID [39, 40, 45, 46]. Besides, picky eating has been hypothesized to be a transdiagnostic indicator of psychopathology in children, as it is associated with high emotional lability, cognitive rigidity, and concurrent symptoms of anxiety and depression [8, 47, 48]. A recent study showed that picky eating is associated with the symptoms of many concurrent psychopathologies in children, and the basis of this relationship is its association with OCD [49]. The present study also had a stronger relationship between picky eating and OCD symptoms compared to other RCADS subscales. This association may be clinically valuable in improving diagnosis and evaluation in the OCD group, which was generally diagnosed lately. Notably, there was also a stronger association between NIAS-fear and RCADS panic disorder compared to other NIAS/RCADS subscales. Although the association between fear-ARFID and panic disorder is poorly understood, there is evidence of symptom and functional overlap between the diagnoses. In the adult NIAS validation sample, there was a strong correlation between NIAS-fear and a measure of visceral sensitivity analogous to anxiety sensitivity in panic disorder [7].

However, it is noteworthy that the GAD scores of RCADS did not correlate with picky eating and appetite scale scores of NIAS. This lack of correlation may be attributed to the complex and heterogeneous presentation of ARFID, which often includes multiple physical symptoms and comorbid psychiatric [6, 50]. Furthermore, the lack of correlation may also be influenced by the unique eating behaviors and attitudes associated with ARFID. Individuals with ARFID may exhibit selective eating patterns, food avoidance based on sensory characteristics, and limited interest in eating, which may not be fully captured by the GAD scores of the RCADS [3, 51]. Future research should continue to explore the specific relationships between anxiety symptoms, eating behaviors, and comorbidities in individuals with ARFID to better understand the lack of correlation observed in this context. The NIAS subscales were diversely correlated with comorbid psychopathology symptoms, highlighting the importance of comorbidity in children with ARFID. Fink et al.'s study [52] at the gastroenterology clinic found that anxiety and depression scores were higher in cases with more severe ARFID symptoms screened by NIAS. The contribution of this data to clinical practice may be for the use of mirtazapine in treatment. Evidence that mirtazapine, which indicates treating adult anxiety and depression, can be used to treat ARFID is increasing daily [53-55]. Although mirtazapine contributes to weight gain by increasing appetite, it may facilitate the treatment of ARFID by treating anxiety and depression, which are often comorbid with ARFID. It may even support cognitive behavioral therapy (CBT) interventions used in treating ARFID. Similarly, a recent study has shown that selective serotonin reuptake inhibitors (SSRIs) and/or hydroxyzine show promise by reducing anxiety in treating ARFID [56].

To the best of our knowledge, this is the first study to investigate the relationship between ARFID symptoms and anxiety, depression, OCD symptoms, and eating behaviors through the NIAS, in addition to the validity study of the Turkish NIAS scale. The results of the study are significant because the study provides a first step towards providing tools to assist in the assessment of ARFID symptoms in young children aged 6-12 years and shows that the NIAS-parent version is a powerful measurement tool in the evaluation of symptoms in Turkish children. The limitations of the study include cross-sectional design, geographic limitations, and data being obtained from a non-clinical sample. In addition, since the age of the children is not suitable for filling out the scales, the parents' filling in the scales may create a bias. This study obtained the children's height and weight based on parental reporting. Studies have demonstrated that the inaccuracy in parents' reporting of their children's height and weight is generally due to underreporting, although it varies depending on the child's age and the country [57]. Studies need to be conducted to calculate the NIAS cut-off value for ARFID, including clinical samples.

Conclusion

This study shows that the Turkish version of the NIAS is valid and reliable in evaluating ARFID symptoms in children. The fact that three subscales of NIAS reflect ARFID presentations in correlation with DSM-5 makes NIAS more advantageous than other scales. In the psychometric properties of the Turkish NIAS, appetite, picky eating, and fear subscales were associated with anxiety and depression symptoms and food-avoidant eating behavior. Picky eating also had a relationship with OCD symptoms. It was also found that ARFID symptoms were not associated with eating disorder symptoms.

As the awareness of ARFID increases daily, the demand for assessment tools also increases. Longitudinal studies that may contribute to the diagnosis and treatment of ARFID will contribute to the clinical progress of patients. For this reason, there is a need for scales and validity studies that have been developed and tested and can support clinical decision-making.

Acknowledgements

We thank the children who participated as volunteers in this study, as well as their families.

Author contributions

HÖ: Conceptualization; data curation; formal analysis; investigation; methodology; project administration; resources; software; supervision; visualization; writing—original draft; writing—review and editing. MK: Conceptualization; data curation; formal analysis; investigation; methodology; project administration; resources; software; visualization; writing—original draft; writing—review and editing. UD: Conceptualization; data curation; formal analysis; investigation; writing—original draft, writing—tion; methodology; validation; visualization; writing—original draft. HFZ: Conceptualization; formal analysis; investigation; methodology; supervision; validation; visualization; writing—original draft, writing. MHT: Conceptualization; formal analysis; investigation; methodology; supervision; sion; validation; visualization; writing—original draft; writing—review and editing.

Funding

There are no funders to report for this submission.

Data availability

The data that support the findings of this study are available from the corresponding author upon reasonable request.

Declarations

Ethical approval and consent to participate

Ethics committee approval was obtained for the study from the Social Sciences University of Ankara Institutional Ethics Committee of Social Sciences and Humanities Research and Publication (No:2022/44352, Date: 08/08/2022).

Competing interests

The authors declare no potential conflict of interest.

Consent for publication

The authors consent for this manuscript to be published. There are no other parties involved in the decision to publish.

Author details

¹Department of Child and Adolescent Psychiatry, Cognitive Behavioral Psychotherapies Association, Karum Is Merkezi Iran Caddesi No: 21 Gaziosmanpasa Mah., 06680 Cankaya, Ankara, Turkey. ²Department of Child and Adolescent Psychiatry, Ankara City Hospital, Ankara, Turkey. ³Departman of Guidance and Counseling, Muğla Sıtkı Koçman University, Muğla, Turkey. ⁴Department of Psychology, University of South Alabama, Mobile, AL, USA. ⁵Department of Pediatrics, Emory University, Atlanta, GA, USA. ⁶Rogers Behavioral Health, Oconomowoc, WI, USA. ⁷Department of Psychology, Social Sciences University of Ankara, Ankara, Turkey.

Received: 28 September 2023 Accepted: 10 February 2024 Published online: 19 February 2024

References

- American Psychiatric Association. American Psychiatric Association. Diagnostic and statistical manual of mental disorders. Washington DC 2013. https://doi.org/10.1176/appi.books.9780890425596.744053.
- Bryant-Waugh R, Kreipe RE. Avoidant/restrictive food intake disorder in DSM-5. Psychiatr Ann. 2012;42:402–5.
- Thomas JJ, Lawson EA, Micali N, Misra M, Deckersbach T, Eddy KT. Avoidant/restrictive food intake disorder: a three-dimensional model of neurobiology with implications for etiology and treatment. Curr Psychiatry Rep. 2017;19:1–9.
- Gonçalves S, Vieira Al, Machado BC, Costa R, Pinheiro J, Conceiçao E. Avoidant/restrictive food intake disorder symptoms in children: Associations with child and family variables. Children's Health Care. 2019;48:301–13.
- Savage JS, Fisher JO, Birch LL. Parental influence on eating behavior: conception to adolescence. J Law Med Ethics. 2007;35:22–34.
- Cooney M, Lieberman M, Guimond T, Katzman DK. Clinical and psychological features of children and adolescents diagnosed with avoidant/ restrictive food intake disorder in a pediatric tertiary care eating disorder program: a descriptive study. J Eat Disord. 2018;6:1–8.
- Zickgraf HF, Ellis JM. Initial validation of the Nine Item Avoidant/Restrictive Food Intake disorder screen (NIAS): A measure of three restrictive eating patterns. Appetite. 2018;123:32–42.
- Zickgraf HF, Richard E, Zucker NL, Wallace GL. Rigidity and sensory sensitivity: Independent contributions to selective eating in children, adolescents, and young adults. J Clin Child Adolesc Psychol. 2022;51:675–87.
- Zickgraf HF, Elkins A. Sensory sensitivity mediates the relationship between anxiety and picky eating in children/adolescents ages 8–17, and in college undergraduates: a replication and age-upward extension. Appetite. 2018;128:333–9.
- He J, Zickgraf HF, Ellis JM, Lin Z, Fan X. Chinese version of the nine item ARFID screen: psychometric properties and cross-cultural measurement invariance. Assessment. 2021;28:537–50.
- Gideon N, Hawkes N, Mond J, Saunders R, Tchanturia K, Serpell L. Development and psychometric validation of the EDE-QS, a 12 item short form of the Eating Disorder Examination Questionnaire (EDE-Q). PLoS ONE. 2016;11:e0152744.
- 12. Fairburn CG, Beglin SJ. Eating disorder examination questionnaire. Cogn Behav Ther Eat Disord. 2008;309:313.
- Esin K, Ayyıldız F. Validity and reliability of the Turkish version of the Eating Disorder Examination Questionnaire (EDE-Q-13): short-form of EDE-Q. J Eat Disord. 2022;10:1–9.
- Wardle J, Guthrie CA, Sanderson S, Rapoport L. Development of the children's eating behaviour questionnaire. J Child Psychol Psychiatry All Discipl. 2001;42:963–70.
- Yılmaz R, Esmeray H, Erkorkmaz Ü. Çocuklarda Yeme Davranışı Anketinin Türkçe uyarlama çalışması. Anatol J Psychiatry/Anadolu Psikiyatri Dergisi. 2011;12:287–94.
- Chorpita BF, Yim L, Moffitt C, Umemoto LA, Francis SE. Assessment of symptoms of DSM-IV anxiety and depression in children: a revised child anxiety and depression scale. Behav Res Ther. 2000;38:835–55.
- Gormez V, Kılınçaslan A, Orengul AC, Ebesutani C, Kaya I, Ceri V, et al. Psychometric properties of the Turkish version of the Revised Child Anxiety and Depression Scale-Child Version in a clinical sample. Psychiatry Clin Psychopharmacol. 2017;27:84–92.
- Neyzi O, Furman A, Bundak R, Gunoz H, Darendeliler F, Bas F. Growth references for Turkish children aged 6 to 18 years. Acta Paediatr. 2006;95:1635–41.
- 19. Team J. JASP (version 0.14. 1)[computer software] 2020.
- Team RC. R: A Language and Environment for Statistical Computing (R version 3.6. 0 (2019-04-26))[Computer software]. The R Project for Statistical Computing 2019.
- Revelle W. Psych: Procedures for psychological, psychometric, and personality research. [R package]. Retrieved from [Left Angle Bracket] https:// CRAN.R-project.org/package=psych [Right Angle Bracket] 2019.
- 22. Rosseel Y, Jorgensen TD, Oberski D, Byrnes J, Vanbrabant L, Savalei V, et al. lavaan: Latent variable analysis. [R package] 2018.
- Epskamp S, Epskamp MS, MplusAutomation S. Package 'semPlot.' Recuperado de https://cran.r-project.org/web/packages/semPlot.pdf 2017.

- Brown TA. Confirmatory factor analysis for applied research. Guilford publications; 2015.
- Schermelleh-Engel K, Moosbrugger H, Müller H. Evaluating the fit of structural equation models: tests of significance and descriptive goodness-of-fit measures. Methods Psychol Res Online. 2003;8:23–74.
- Keith TZ. Multiple regression and beyond: an introduction to multiple regression and structural equation modeling. Milton Park: Routledge; 2019.
- 27. Kline RB. Principles and practice of structural equation modeling. New York: Guilford publications; 2015.
- Özdamar K. Paket Programlar ile İstatistiksel Veri Analizi Kaan Kitabevi. Baskı, Eskişehir 2004:528–673.
- 29. Ziółkowska B, Ocalewski J, Zickgraf H, Brytek-Matera A. The Polish Version of the Avoidant/Restrictive Food Intake Disorder Questionnaire—Parents Report (ARFID-Q-PR) and the Nine Items Avoidant/Restrictive Food Intake Disorder Screen—Parents Report (NIAS-PR): Maternal Perspective. Nutrients. 2022;14:3175.
- Medina-Tepal KA, Vazquez-Arevalo R, Trujillo-ChiVacuán EM, Zickgraf HF, Mancilla-Díaz JM. Cross-cultural adaptation and validation of the Nine Item ARFID Screen (NIAS) in Mexican youths. Int J Eat Disord. 2023;56(4):721–6.
- Burton Murray H, Dreier MJ, Zickgraf HF, Becker KR, Breithaupt L, Eddy KT, et al. Validation of the nine item ARFID screen (NIAS) subscales for distinguishing ARFID presentations and screening for ARFID. Int J Eat Disord. 2021;54:1782–92.
- Ellis JM, Galloway AT, Webb RM, Martz DM. Measuring adult picky eating: the development of a multidimensional self-report instrument. Psychol Assess. 2017;29:955.
- Hunot C, Fildes A, Croker H, Llewellyn CH, Wardle J, Beeken RJ. Appetitive traits and relationships with BMI in adults: development of the Adult Eating Behaviour Questionnaire. Appetite. 2016;105:356–63.
- 34. Mallan KM, Fildes A, de la Piedad GX, Drzezdzon J, Sampson M, Llewellyn C. Appetitive traits associated with higher and lower body mass index: evaluating the validity of the adult eating behaviour questionnaire in an Australian sample. Int J Behav Nutr Phys Act. 2017;14:1–8.
- Brown CL, Vander Schaaf EB, Cohen GM, Irby MB, Skelton JA. Association of picky eating and food neophobia with weight: a systematic review. Child Obes. 2016;12:247–62.
- Faith MS, Scanlon KS, Birch LL, Francis LA, Sherry B. Parent-child feeding strategies and their relationships to child eating and weight status. Obes Res. 2004;12:1711–22.
- Taylor CM, Wernimont SM, Northstone K, Emmett PM. Picky/fussy eating in children: review of definitions, assessment, prevalence and dietary intakes. Appetite. 2015;95:349–59.
- Smith BL, Gutierrez R, Ludlow AK. A comparison of food avoidant behaviours and sensory sensitivity in adults with and without Tourette syndrome. Appetite. 2022;168:105713.
- Zickgraf HF, Lane-Loney S, Essayli JH, Ornstein RM. Further support for diagnostically meaningful ARFID symptom presentations in an adolescent medicine partial hospitalization program. Int J Eat Disord. 2019;52:402–9.
- Zickgraf HF, Murray HB, Kratz HE, Franklin ME. Characteristics of outpatients diagnosed with the selective/neophobic presentation of avoidant/ restrictive food intake disorder. Int J Eat Disord. 2019;52:367–77.
- Bryson AE, Scipioni AM, Ornstein RM. Avoidant/restrictive food intake disorder: categorization of subtypes and longitudinal outcomes after day hospitalization. J Adolesc Health. 2017;60:S45–6.
- 42. Nicely TA, Lane-Loney S, Masciulli E, Hollenbeak CS, Ornstein RM. Prevalence and characteristics of avoidant/restrictive food intake disorder in a cohort of young patients in day treatment for eating disorders. J Eat Disord. 2014;2:1–8.
- Fisher MM, Rosen DS, Ornstein RM, Mammel KA, Katzman DK, Rome ES, et al. Characteristics of avoidant/restrictive food intake disorder in children and adolescents: a "new disorder" in DSM-5. J Adolesc Health. 2014;55:49–52.
- Duncombe Lowe K, Barnes TL, Martell C, Keery H, Eckhardt S, Peterson CB, et al. Youth with avoidant/restrictive food intake disorder: examining differences by age, weight status, and symptom duration. Nutrients. 2019;11:1955.

- Norris ML, Spettigue W, Hammond NG, Katzman DK, Zucker N, Yelle K, et al. Building evidence for the use of descriptive subtypes in youth with avoidant restrictive food intake disorder. Int J Eat Disord. 2018;51:170–3.
- Reilly EE, Brown TA, Gray EK, Kaye WH, Menzel JE. Exploring the cooccurrence of behavioural phenotypes for avoidant/restrictive food intake disorder in a partial hospitalization sample. Eur Eat Disord Rev. 2019;27:429–35.
- Zucker N, Copeland W, Franz L, Carpenter K, Keeling L, Angold A, et al. Psychological and psychosocial impairment in preschoolers with selective eating. Pediatrics. 2015;136:e582–90.
- Fernandez C, McCaffery H, Miller AL, Kaciroti N, Lumeng JC, Pesch MH. Trajectories of picky eating in low-income US children. Pediatrics. 2020;145:e20192018.
- Schwarzlose RF, Hennefield L, Hoyniak CP, Luby JL, Gilbert KE. Picky eating in childhood: associations with obsessive-compulsive symptoms. J Pediatr Psychol. 2022;47:816–26.
- Datta N, Matheson B, Walker AC, Van Wye E, Lock JD. Psychoeducational and motivational treatment for low-weight Avoidant/Restrictive Food Intake Disorder (ARFID): three case reports in school-aged children. Clin Child Psychol Psychiatry. 2023. https://doi.org/10.1177/135910452311691 41.
- 51. Zickgraf HF, Franklin ME, Rozin P. Adult picky eaters with symptoms of avoidant/restrictive food intake disorder: comparable distress and comorbidity but different eating behaviors compared to those with disordered eating symptoms. J Eat Disord. 2016;4:1–11.
- Fink M, Simons M, Tomasino K, Pandit A, Taft T. When is patient behavior indicative of avoidant restrictive food intake disorder (ARFID) vs reasonable response to digestive disease? Clin Gastroenterol Hepatol. 2022;20:1241–50.
- Naviaux A-F. Management of ARFID (Avoidant Restrictive Food Intake Disorder) in a 12-year-old on a paediatric ward in a general hospital: Use of mirtazapine, partial hospitalisation model and family based therapy. Psychiatr Danub. 2019;31:421–6.
- Naguy A, Roshdy R, Al-Mutairi A, Alwetayan S, Alamiri B. Mirtazapine improved eating patterns in avoidant/restrictive food intake disorder. Am J Ther. 2022;30:173–5.
- Gray E, Chen T, Menzel J, Schwartz T, Kaye WH. Mirtazapine and weight gain in avoidant and restrictive food intake disorder. J Am Acad Child Adolesc Psychiatry. 2018;57:288–9.
- Mahr F, Billman M, Essayli JH, Lane Loney SE. selective serotonin reuptake inhibitors and hydroxyzine in the treatment of avoidant/restrictive food intake disorder in children and adolescents: rationale and evidence. J Child Adolesc Psychopharmacol. 2022;32:117–21.
- 57. Chai LK, Collins CE, May C, Holder C, Burrows TL. Accuracy of parentreported child height and weight and calculated body mass index compared with objectively measured anthropometrics: Secondary analysis of a randomized controlled trial. J Med Internet Res. 2019;21:e12532.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.